Author's response to reviews

Title: HER-2, p53, p21 and hormonal receptors proteins expression as predictive factors of response and prognosis in locally advanced breast cancer treated with neoadjuvant docetaxel plus epirubicin combination

Authors:

Daniel G Tiezzi (daniel.tiezzi@yahoo.com)
Jurandyr M Andrade (andrade@fmrp.usp.br)
Alfredo Ribeiro-Silva (arsilva@fmrp.usp.br)
Fabio E Zola (fzola@hotmail.com)
Heitor RC Marana (marana@keynet.com.br)
Marcelo G Tiezzi (marcelotiezzi@uol.com.br)

Version: 4 Date: 14 February 2007

Author's response to reviews: see over
Author's response to reviewer

Title: HER-2, p53, p21 and hormonal receptors proteins expression as predictive factors of response and prognosis in locally advanced breast cancer treated with neoadjuvant docetaxel plus epirubicin combination

Authors:
Daniel G. Tiezzi: daniel.tiezzi@yahoo.com
Jurandyr M. Andrade: andrade@fmrp.usp.br
Alfredo Ribeiro-Silva: arsilva@fmrp.usp.br
Fabio E. Zola: fabiozola@hotmail.com
Heitor R. C. Marana: maran@keynet.com.br
Marcelo G. Tiezzi: marcelotiezzi@uol.com.br

Version: 3 Date: 6 February 2007

Reviewer: Sehwan Han

Reviewer's report:

General

1. The part which I pointed were well mentioned. The "discussion" seems still redundant.

I could not realize the redundancy in "Discussion". The "discussion" structure seems clear. However, I made the necessary corrections trimming superfluous points.

2. Mistype was detected on the first paragraph of the "Discussion", citotoxic should be revised as 'cytotoxic'.

Done.